Author's response to reviews

Title: Improved metabolic health among the obese in six population surveys 1986 to 2009. The Northern Sweden MONICA study.

Authors:

Martin Benckert (martin.benckert@gmail.com)
Mikael Lilja (mikael.lilja@jll.se)
Stefan Söderberg (stefan.soderberg@medicin.umu.se)
Mats Eliasson (mats.eliasson@nll.se)

Version: 2 Date: 14 September 2014

Author's response to reviews:

This version now includes line and page numbers